

Make Your Publications Visible.

A Service of



Leibniz-Informationszentrum Wirtschaft Leibniz Information Centre

Githaiga, Peter Nderitu; Kilong'i, Andrew Wenani

#### **Article**

Foreign capital flow, institutional quality and human capital development in sub-Saharan Africa

**Cogent Economics & Finance** 

#### **Provided in Cooperation with:**

**Taylor & Francis Group** 

Suggested Citation: Githaiga, Peter Nderitu; Kilong'i, Andrew Wenani (2023): Foreign capital flow, institutional quality and human capital development in sub-Saharan Africa, Cogent Economics & Finance, ISSN 2332-2039, Taylor & Francis, Abingdon, Vol. 11, Iss. 1, pp. 1-18, https://doi.org/10.1080/23322039.2022.2162689

This Version is available at: https://hdl.handle.net/10419/303929

#### Standard-Nutzungsbedingungen:

Die Dokumente auf EconStor dürfen zu eigenen wissenschaftlichen Zwecken und zum Privatgebrauch gespeichert und kopiert werden.

Sie dürfen die Dokumente nicht für öffentliche oder kommerzielle Zwecke vervielfältigen, öffentlich ausstellen, öffentlich zugänglich machen, vertreiben oder anderweitig nutzen.

Sofern die Verfasser die Dokumente unter Open-Content-Lizenzen (insbesondere CC-Lizenzen) zur Verfügung gestellt haben sollten, gelten abweichend von diesen Nutzungsbedingungen die in der dort genannten Lizenz gewährten Nutzungsrechte.



https://creativecommons.org/licenses/by/4.0/

#### Terms of use:

Documents in EconStor may be saved and copied for your personal and scholarly purposes.

You are not to copy documents for public or commercial purposes, to exhibit the documents publicly, to make them publicly available on the internet, or to distribute or otherwise use the documents in public.

If the documents have been made available under an Open Content Licence (especially Creative Commons Licences), you may exercise further usage rights as specified in the indicated licence.





### **Cogent Economics & Finance**



ISSN: (Print) (Online) Journal homepage: www.tandfonline.com/journals/oaef20

## Foreign capital flow, institutional quality and human capital development in sub-Saharan Africa

Peter Nderitu Githaiga & Andrew Wenani Kilong'i

**To cite this article:** Peter Nderitu Githaiga & Andrew Wenani Kilong'i (2023) Foreign capital flow, institutional quality and human capital development in sub-Saharan Africa, Cogent Economics & Finance, 11:1, 2162689, DOI: 10.1080/23322039.2022.2162689

To link to this article: <a href="https://doi.org/10.1080/23322039.2022.2162689">https://doi.org/10.1080/23322039.2022.2162689</a>

9	© 2023 The Author(s). This open access article is distributed under a Creative Commons Attribution (CC-BY) 4.0 license.
	Published online: 31 Dec 2022.
	Submit your article to this journal 🗹
hil	Article views: 2914
Q	View related articles ☑
CrossMark	View Crossmark data ☑
4	Citing articles: 5 View citing articles 🗗







Received: 06 June 2022 Accepted: 21 December 2022

\*Corresponding author: Peter Nderitu Githaiga, Department of Accounting and Finance, Moi University, Eldoret, Kenva

E-mail: nderitugithaiga@mu.ac.ke

Reviewing editor: David McMillan, University of Stirling, Stirling, UK

Additional information is available at the end of the article

#### FINANCIAL ECONOMICS | RESEARCH ARTICLE

# Foreign capital flow, institutional quality and human capital development in sub-Saharan Africa

Peter Nderitu Githaiga<sup>1\*</sup> and Andrew Wenani Kilong'i<sup>2</sup>

**Abstract:** This study investigates the moderating effect of institutional quality on the relationship between foreign capital flow and human capital development in sub-Saharan Africa. The study uses a sample of 34 countries in sub-Saharan Africa and data for 2009 to 2019. Human capital development is measured using the Human Development Index (HDI). To control for endogeneity, the study uses the system generalized method of moments (GMM) estimator. The results demonstrate a positive relationship between remittances, foreign direct investment (FDI), institutional quality and human capital development. Official development assistance (ODA), on the other hand, has a negative and significant effect on human capital development. The findings further reveal that the effect of remittances and FDI on the human capital development is moderated by the institution's quality. However, the effect of ODA on the development of human capital is not influenced by institutional quality. Findings from the study provide valuable insights to policymakers. This study highlights the importance of remittances and FDI in stimulating human capital development in sub-Saharan Africa. Additionally, the study reveals the harmful impact of official development on human capital development that necessitates policy interventions. Drawing on these findings, policymakers should undertake policy reforms to improve the quality of institutions and enhance the impact of foreign capital flows on human development. This study offers two contributions. First, the study fills a vacuum in the literature by focusing on the relationship between foreign capital flow, institutional quality, and the development of human capital in sub-Saharan Africa. Second, the SSA is one of the developing nations that has seen a significant brain drain because of widespread migration to industrialized nations. Therefore, it is necessary to investigate how much foreign capital flows through the development of human capital contribute to the socioeconomic change of the region.

Subjects: Culture & Development; Economics and Development; Economics; Finance; Business, Management and Accounting

Keywords: foreign capital flow; human capital development; Sub-Saharan Africa; panel data

#### 1. Introduction

According to theory (Lucas, 1988; Romer, 1990a), building human capital is essential to fostering economic progress. Many developing nations have exploited foreign capital flows to boost human







capital, which ultimately leads to economic growth and all-around development. The most pervasive idea among researchers and policymakers is that foreign capital flow enhances growth in diverse ways by increasing the capital base of receiving countries and providing employment opportunities. Foreign capital, such as FDI, further brings technological development to local firms, which improves the productivity of host countries and stimulates economic growth. Therefore, foreign capital flows have been instrumental in fostering socio-economic development and complementing national development initiatives in the recipient countries. Sub-Saharan Africa, like other developing regions, depends on foreign funding to finance projects and achieve socio-economic growth (Adams et al., 2017).

The main components of foreign capital flows are remittances, foreign direct investment and official development assistance (Musibau et al., 2019). Remittances are the second-largest source of external finance after foreign direct investment in the developing economies (Bhattacharya et al., 2018). As of 2021, the estimated amount of remittance flow to developing countries was \$589 billion in 2021 (Word Bank, 2021). Remittances are goods and money sent by individuals working outside of their country of origin to their family members back home (Kulkami, 2014). These transfers are made informally by hand as cash or in-kind, through the financial system, through money transfer services or through the banking system. Remittances are often intended for use by families back home and are not intended for investment (Heyse et al., 2015). Altruism, self-interest, asset accumulation, and insurance are a few of the reasons Lucas and Stark (1985) provide for sending remittances. Migrants have an altruistic concern for the welfare of those who are left behind. The self-interest incentive is centered on migrants' desire to purchase real estate at home. Remittances and migration are regarded as a type of risk-sharing or risk-diversification for insurance purposes (Lucas & Stark, 1985).

The OECD defines FDI as an investment that "reflects the objective of obtaining a lasting interest by a resident entity in one economy ('direct investor') in an entity resident in an economy other than that of the investor ('direct investment enterprise')" (OECD, 2005). FDI is the largest component of foreign capital flow available to developing nations (Katoka & Kwon, 2018). By 2021, the estimated amount of FDI flow to developing countries is \$616 billion, accounting for over 75% of total global private capital flow (UNCTAD, 2021). FDI is in two forms, horizontal and vertical. Horizontal FDI is characterized by duplicating production capacity in the host country, being practically every market served by its power. Vertical FDI involves transferring production capacity from the country of origin to the host country to increase efficiency (Munteanu, 2015). Additionally, FDI is the most significant method for transferring foreign technology across nations, along with international trade (Zamani & Tayebi, 2022).

Foreign grant-in aid generally flows in the form of official development assistance (ODA) from developed countries to developing countries (OECD, 2012). The estimated amount of development assistance to developing countries is USD 161.2 billion (OECD, 2021). Official development assistance aims to promote the provision of public goods and services among the least developed and developing countries. ODA supplements other local or foreign sources of finance that support infrastructural development. ODA can also be in the form of technical assistance and capacity building of institutions. Therefore, aid to the developing countries is tailored to bring forth economic growth, reduce poverty and better living standards.

Human capital development is the key to economic growth and the realization of sustainable development goals (Barua, 2021; Idike et al., 2021; Matousek & Tzeremes, 2021). Extant literature shows that scholars have defined human capital differently. Mincer (1996) views human capital development as an indicator of the proportion of a country's population with sufficient education, skills, and training. Torruam et al. (2014) suggest that human capital development involves developing skills, knowledge, productivity, and inventiveness of human capital formation. However, Romer (1990a) defines human capital development as "skills acquisition through training of workers as well as investment in health-care reforms." Meanwhile, Jhinger (2005) suggests that



expenditure on human capital development necessitates investment in education, health, and social services. Studies also point to health care as an additional dimension of human capital development (Buevich et al., 2019; Romer, 1990b). The United Nations Development Programme (UNDP) models a nation's human capital from three perspectives: school enrolment and literacy, the standard of living, and life expectancy. Similarly, a Human Development Report by the United Nations Development Program (UNDP; 2018) suggests that countries must invest in societies' education, nutrition, health, and employment skills to realize economic growth. Consequently, human capital development is an indicator of a country's investment in education and training, nutrition and health care to improve human productivity and economic growth.

Proponents of the Human Capital Theory posit that a nation's socio-economic development centers on continued investment in the education and health of its population (Affandi et al., 2019; Schultz, 1961). Therefore, no country can overlook the importance of human capital development in socio-economic development. Previous studies have also revealed that a country with a highly educated and healthier population produces more output and ultimately improves economic growth (Bloom et al., 2019; Yao, 2019). Since immemorial educational attainment, nutrition and health have long been hypothesized as key drivers of labour productivity. According to Petty (1769), raising the value of a nation's human capital promotes economic growth. At the same time, Smith (1904) associated national prosperity with an increased division of labour, advocated for higher wages, and enhanced workers' skills. As opposed to the conventional wisdom that physical and tangible assets like factories, capital, machinery, and land are the primary drivers of economic growth, contemporary economists assert that countries accomplish economic growth by investing in the development of their human capital (Penrose, 1959; Weisbrod, 1962).

Because of the vital relationship between human capital development and socio-economic development, there have been numerous research studies on the determinants of human capital development, particularly in developing countries. Although prior studies claim that budgetary constraints explain the low level of human capital development in developing countries (Olopade et al., 2019; Wenzel, 2017), these countries receive a considerable amount of foreign capital inflows; foreign remittances, official development assistance and foreign direct investment (Ahmed et al., 2021; Wang et al., 2022). Several studies have examined the relationship between foreign capital inflows and human development, but the findings are inconclusive (Kheng et al., 2017; Ponce et al.,, 2008; Kroeger & Anderson, 2014). Besides, none of these studies examined the foreign capital flow and human capital development relationship in sub-Saharan Africa, which is considered among regions with the lowest level of human capital development (Matousek & Tzeremes, 2021). SSA Africa has been of much interest in development studies since many of the countries face a lack of strong legal and political institutions: lack of property rights, weak courts and deficiency in contract enforcement, violence and political instability, hostile regulatory environment and high inflation (Asongu & Nwachukwu, 2016; Sarwar et al., 2015). In addition, despite receiving a fair share of foreign capital flow, countries within SSA fall between the low and medium HDI range (Darley, 2012)

Prior studies show suggest that dimensions of institutional qualities such as corruption, political stability, rule of law, violence and terrorism affect human capital development (Borja, 2020; Ortega et al., 2016). Additionally, previous studies also reveal that foreign capital flows are more effective in countries with sound institutions. In their paper, Collier and Dollar (2002) note that the effectiveness of foreign aid is determined by the quality of policies and institutions. In the same vein, a study by Kosack (2003) found that the effectiveness of foreign aid in improving the quality of life is enhanced by democratization of institutions. Overall, it has been found that aid empowers the poor in a good institutional setting and improves the functioning and capabilities of individuals. A study by Kathavate and Mallik (2012) reports the effect of aid on the growth depends on the quality of recipient country's institutions. Using a sample of 90 developing and transition countries for the period 1970–2012, Lartey and Mengova (2016) find that institutional quality influences the flow of remittances. Therefore, the quality of political institutions may also influence the flow of



foreign capital. Similarly, Ajide and Raheem (2016), report that institutional quality plays a crucial role in attracting remittance inflow to the ECOWAS region. The literature also shows that institutional quality is a key determinant of FDI inflow (Buchanan et al., 2012; Shah et al., 2016). Thus, countries with better governance institutions are likely to attract more foreign capital, which in response will positively influence human capital development. However, empirical links between foreign capital flows, institutional quality and human capital development are still limited. To our knowledge, no other study has addressed this issue. From this background, this study seeks to assess whether institutional quality moderates the relationship between foreign capital inflows and human capital development in sub-Saharan Africa. The rest of the paper proceeds as follows: section 2 presents an overview of sub-Saharan Africa. Section 3 deals with the empirical literature. Section 4 presents the research methodology, sample and data. Section 5 discusses the findings, while section 6 concludes.

#### 2. Background of Sub-Saharan Africa

Sub-Saharan Africa (SSA) is geographically the area of the Africa continent that lies south of the Sahara and consists of all African countries and territories. As of 2018, SSA had a GDP per capita (Current US\$) that stood at 1,574.2 compared with 1,905.0 in South Asia, 11,132.2 in developing East Asia and Pacific, 8,057.0 in developing the Middle East and North Africa, 9,023.5 in developing Latin America and the Caribbean, 25,078.0 in developing Europe and Central Asia, and 36,546.4 European Union (World Bank, 2019). By 2019, about 41.4.0% of the sub-Saharan Africa population lived on less than \$1.90 a day, compared to 16.1% South Asia, 4.2% Middle East & North Africa, and 3.9% for Latin America & the Caribbean. It has the highest annual population growth rate of 2.7% for all regions in the world; 1.7% in South Asia, 0.6% East Asia and the Pacific, 1.7% in the Middle East & North Africa, and 0.2% in Europe & Central Asia (World Bank, 2019). Additionally, UNCTAD (2018) classifies 32 SSA countries as least developed. In line with the Human Development Index (HDI) of 2018, the sub-Saharan African region is ranked the lowest with an HDI of 0.537, compared to 0.733 in East Asia, 0.771 in Europe and Central Asia, 0.758 in Latin America and the Caribbean, and 0.638 in South Asia (United Nations Development Program (UNDP), 2018). In addition, the region again records the highest prevalence of undernourishment in the world, reaching as high as 23% undernourished population (FAO and ECA, 2018). The report further shows that out of the 821 million malnourished people globally, 236 million are from sub-Saharan Africa.

A report by the World Health Organization (2006) shows that 72% of the sub-Saharan African population still lacks access to essential sanitation services. Besides, the region is home to 53% of the world's people living with HIV and AIDs (UNAIDS, 2018). In 2019, SSA had a life expectancy of fewer than 61 years, compared to 69 years in South Asia, 78 years in Europe & Central Asia, and 76 in East Asia & the Pacific (World Bank, 2019). Sub-Saharan Africa also faces the lowest literacy level (% of people ages 15 and above) of 66% compared with 72% in South Asia, 98% in Europe & Central Asia and 96% in East Asia & the Pacific. According to the statistics provided above, sub-Saharan Africa is one of the regions with the lowest levels of human capital development. The region does, however, receive a substantial amount of foreign capital transfers in the form of foreign direct investment, migrant remittances, and official development assistance (foreign aid). SSA got almost 88% of the 32.4% of official development assistance allocated to Africa in 2019 (OECD, 2019). Additionally, the region receives 10.2% of worldwide remittances or 2.18% of the region's GDP (World Bank, 2019). Additionally, according to UNCTAD (2018), SSA receives nearly 2.9% of all foreign direct investment in the world (FDI). However, it is still unclear how foreign capital flow and the growth of human capital in sub-Saharan Africa are related.

#### 3. Review of previous studies

#### 3.1. Remittance and human capital development

Empirically, a large part of the literature has focused on how remittance influences development through economic growth. Remittances may spur economic growth by increasing entrepreneurial



activity and investment by alleviating credit constraints, especially in developing countries with less efficient credit markets. A study by Azam and Raza (2016) examined the influence of workers' remittances on human capital development and employed a sample of 17 countries and panel data from 1996 to 2013. The results indicate that workers' remittances positively and significantly affect human capital development. Borja (2020) assessed the relationship between remittances and human development. The study used a sample of 26 Latin American and Caribbean countries. 1985-2016. The findings show a positive relationship between remittances and human capital development. Additionally, the study suggests that corruption constrains the effectiveness of remittances in improving human capital development. Similarly, Umar (2022) examined the relationship between remittances, institutions and human development (HD) in sub-Saharan African (SSA) countries using data from 2004 to 2018. The findings indicate a positive and significant impact of remittances on HD in SSA. The results further reveal a substitutional relationship between institutions and remittances in stimulating HD. The estimations mean that remittances promote HD in countries with a weak institutional environment. The findings also establish that the marginal significance of remittances as a source of capital for HD falls in countries with well-developed institutions. Using data for the period between 2007 and 2010, Salas (2014) also report a positive relationship between remittances and human capital formation in Peru. Hassan et al. (2013) assessed the relationship between remittances and human capital formation in Pakistan for the period between 1981 and 2011. The authors found that worker's remittances had a negative effect on human capital formation. Based on the literature, this study hypothesizes that,

#### H1. Remittances positively affects human capital development

#### 3.2. Foreign direct investment and human capital development

Many academics have shown solid evidence that FDI stimulates socio-economic development, particularly in developing countries with sound institutional and political environments (Burnside & Dollar, 1997; Sethi et al., 2019; Acemoglu et al., 2005; La Porta et al., 1997). Studies also confirm that foreign investment may advance human capital in poor and emerging economies (Suliman & Mollick, 2009; Zhuang, 2017). For instance, a study by Kar (2013) that sought to examine the relationship between foreign direct investment (FDI) and human capital development (expenditure on higher education) in India reported a weak unidirectional causality between FDI and human capital development. Kaulihowa and Adjasi (2019) assessed the non-linear effect of FDI and the different levels of human capital development. The study used a sample of nine African countries over 2000-2017. The study found mixed findings. While the effect of FDI on primary education was an inverted U-shaped relationship, tertiary education showed a U-shaped pattern. However, the FDI and secondary education association is insignificant. Henok and Kaulihowa (2021) investigated the impact of FDI on human capital development in Southern African Customs Union (SACU) countries. The study used a sample of five countries and data from 1990 to 2018. The findings indicate that FDI improves human capital development when the primary school enrolment rate is used. Conversely, the FDI and secondary education relationship is negative. Zhuang (2017) examined the effect of inward FDI on human capital accumulation. The study employed a panel of sample data from 16 East Asian countries from 1985 to 2010. The findings show that an increase in foreign presence is associated with an increase in secondary schooling. However, the relationship between FDI and tertiary schooling was negative. Kheng et al. (2017) assessed the human capital and foreign direct investment (FDI) relationship using panel data from 55 developing countries from 1980 to 2011. The results show a significant bi-directional causality between human capital and FDI. Fagbemi and Osinubi (2020) investigated the interconnections between FDI and human capital development in Nigeria. The study used data for the period 1981-2018 and was analyzed through nonlinear autoregressive distributed lag (NARDL) and linear ARDL. The results confirm that the effect of FDI on human capital development was insignificant in the long run but significant in the short run. Prior studies report that the recipient country's level of human capital development determines the amount of FDI received and its absorptive capacity. Kheng et al. (2017) examined the relationship between human capital and foreign direct investment. The authors used panel data from 55 developing countries



for the period 1980–2011. The results reveal a significant bi-directional causality between human capital and FDI. Therefore, this study postulates that,

H2. Foreign direct investment positively affects human capital development

#### 3.3. Official development assistance and human capital development

Asiama and Quartey (2009) assessed the impact of foreign aid on human development. The study used data drawn from 39 SSA countries for the year 2005. The study shows no relationship between foreign aid and human development indicators. Yiheyis and Woldemariam (2020) investigated the association between remittance, official development assistance and human development. The study used a sample of 35 African countries. The study findings reveal a positive relationship between official development assistance, remittances and human capital development. Shirazi et al. (2009) assessed the relationship between foreign aid and human capital development using time-series data for 5 years. The findings show a positive relationship between foreign aid and human development. Moe (2008) investigated the relationship between official development assistance (ODA) and human and educational developments among Southeast Asian countries and data from 1990 to 2004. The study found that foreign direct investment and official development assistance positively and significantly influenced human development. Using data for the period 2006 to 2016 and a sample of 15 Asian countries, Lee et al. (2019), found a positive association between ODA and human capital development. S. A. Asongu and Tchamyou (2019) assessed the relationship between foreign aid, education and lifelong learning. The study considered a sample of 53 African countries and data for the period 1996–2010. The findings reveal that foreign aid positively affected primary school enrolment and lifelong learning. In addition, the study found that the effect of aid dynamics on secondary and tertiary school enrolments is not significant. Yogo (2017) assessed aid effectiveness and its efficient use in achieving universal primary education in sub-Saharan Africa. A sample of 35 sub-Saharan countries over the period 2000 to 2010. The results of this study show that foreign aid significantly increased the primary school completion rate. Conversely, some studies have reported a negative association between ODA and human capital development. Lohani (2004) who considered a sample of 120 developing countries reports a negative relationship between foreign aid and human development. Using data from a sample of 22 African countries for the period 1996–2009, Asongu (2014) found a negative association between foreign aid and human capital development. Based on the empirical literature, this study hypothesizes that,

H3. Official development assistance positively affects human capital development

#### 3.4. Foreign capital flow, institutional quality, human capital development

Low quality of institutions of governance deters human capital development as vital resources are diverted toward less efficient uses. The institutional theory put forward by North (1990) offers an important theoretical insight into the significance of institutions in attracting foreign capital flows and fostering socioeconomic development. North (1990) argues that while strong institutions increase economic activity by lowering transaction costs and output costs, weak and deficient institutions impede productivity and increase uncertainty. Effective political and legal structures, particularly the protection of property rights and the rule of law, are reportedly necessary for resource allocation (Rodrik et al., 2004). Investment costs are typically high in unfavorable institutional environments with intellectual property theft, bad governance, and corruption (Daniele & Marani, 2011). Investors and donors, in particular, are hesitant to transfer capital to such economies. Therefore, institutional stability is essential for attracting foreign capital and ultimately fostering the development of human capital. Although there are conflicting and inconclusive findings in the empirical literature, the link between institutional quality and the development of human capital is well documented.



Fagbemi et al. (2022) assessed the effect of corruption on the growth of the human capital in Nigeria from 1996 – 2019. According to the authors, corruption had a significant effect on the growth of human capital. Ouedraogo et al. (2022) studied the association between institutional quality and human capital development. The authors considered 49 African countries and data for the period be 1996–2018, and they found that government effectiveness, the level of corruption, political stability and the violence are key determinants of human capital development. In Brazil, Koppensteiner and Menezes (2021) reported that violence has a negative effect on school attendance, standardized test scores and that it increases the dropout rates of students. In the same line of research, Bertoni et al. (2019) assessed the effect of the Boko Haram conflict on educational outcomes of persons residing in north-eastern Nigeria for the period between 2009 and 2016. The authors found that the conflict reduced school enrollment. Di Maio and Nandi (2013) looked studied how the Israeli-Palestinian conflict affected Palestinian children's access to education and child labor in the West Bank. The finding revealed that the violence reduced the number of students attending school in the area.

Hence, some level of institutional quality is required to leverage foreign financial flows for the development of human capital. For instance, in a nation with a high incidence of corruption, foreign investors might be discouraged from making investments in healthcare and education. Bénassy-Quéré et al. (2007) concluded that the factors that most strongly influence inbound FDI include bureaucracy, corruption, information, the banking industry, and legal institutions using data collected from 52 nations during the years 1985 to 2000. Similarly, Moore and Daday (2010) studied the barriers to human capital development in Swaziland, Cameroon, and Kenya for the period 1990 to 2004. The authors found that the barriers to human capital development were HIV/ AIDS, land tenure issues, and political corruption. Using a sample of 88 developing countries, Borja (2014) found that remittances are effective at promoting economic growth when adequate institutions are set in place. Borja (2017) also reports that FDI and remittances have a more positive impact on economic growth if corruption is controlled. Generally, going by the existing literature there may be a relationship between foreign capital flows, institutional quality and human capital development, but there is no empirical evidence to support it. Thus, this research aims at closing this gap. Moreover, previous works are limited because most studies considered each of the foreign capital flow components separately. The study posits that institutional quality influences the relationship between foreign capital flows and human capital development. To investigate the moderating effect of institutional quality for each of the components of foreign capital flow and human capital development, the study hypotheses as follows:

H4. Institutional quality positively affects the human capital development

H5a. Institutional quality moderates the relationship between remittances and human capital development.

H5b. Institutional quality moderates the relationship between foreign direct investment and human capital development.

H5c. Institutional quality moderates the relationship between official development assistance and human capital development.

#### 4. Methodology

#### 4.1. Sample and data

The study targeted sub-Saharan Africa (SSA) countries between 2009 and 2019. This timeframe is ideal because it lies between the Post-Global Financial Crisis (2007–2008) of and the pre-Covid-19 Pandemic periods. The period also witnessed a relatively stable economic and political stability



that is necessary for human capital development. The sampling framework comprised the 54 countries that constitute sub-Saharan Africa. The inclusion and exclusion criteria were based on whether the selected country had complete data for the period. After data collection, the final sample comprised 41 countries. Panel data were extracted from the World Development Indicators (WDI) by the World Bank and the OECD. The dependent variable employed in this study is human capital development, operationalized by the UNDP's Human Development Index (HDI). The predictor variable is foreign capital inflow, which is disintegrated into foreign direct investment, foreign remittances and official development assistance. Data on the predictor variables are available on the World Bank database, while the definition and source of data for the control variables are shown in Table A1 Appendix.

#### 4.2. Econometric model

This study examines the moderating impact of institutional quality on the relationship between foreign capital flow and human capital development using the system generalized method of moments (GMM) estimator developed by Arellano and Bond (1991), Arellano and Bover (1995), and Blundell and Bond (1998). The ability of S-GMM to handle potential endogeneity justifies the choice. The consistency of the S-GMM estimator relies on two specification tests: (i) a test of the joint validity of all instruments and (ii) the absence of second-order autocorrelation AR(2). The first test is referred to as the Sargan/Hansen's J-test of over-identification and the null hypothesis test, which suggests that instruments are uncorrelated with the error term. The second test examines the hypothesis of no second-order autocorrelation in the error term of the difference equation (Arellano & Bond, 1991). Human capital development is the study's dependent variable, and it is measured by the Human Development Index (HDI). Remittances, official development assistance, and foreign direct investment are the exogenous variables (ODA). Institutional quality is a moderator and is measured by the Country Policy and Institutional Assessment (CPIA). Based on previous empirical studies, the study controls for two macroeconomic factors: population growth rate (Boucekkine et al., 2013; Fertig et al., 2009), as well as government expenditure on health and education (Dissou et al., 2016; Yang, 2020). The regression models for the research are shown below.

Model 1. Testing the direct effect

$$HDI_{it} = \beta_0 + \beta_1 GEEH_{it} + \beta_2 PGR_{it} + \beta_3 REM_{it} + \beta_4 FDI_{it} + \beta_5 ODA_{it} + \epsilon_{it}$$

Model 2. Testing the effect of the moderator on the outcome variable

$$HDI_{it} = \beta_0 + \beta_1 GEEH_{it} + \beta_2 PGR_{it} + \beta_3 REM_{it} + \beta_4 FDI_{it} + \beta_5 ODA_{it} + \beta_6 CPIA_{it} + \epsilon_{it}$$

Model 3-5. Testing the effect of the interaction terms on the outcome variable (stepwise).

$$\begin{aligned} \text{HDI}_{it} &= \beta_0 \ + \beta_1 \text{GEEH}_{it} + \beta_2 \ \text{PGR}_{it} + \beta_3 \text{REM}_{it} + \beta_4 \text{FDI}_{it} + \beta_5 \text{ODA}_{it} + \beta_6 \text{CPIA}_{it} + \beta_7 \text{REM} * \text{CPIA}_{it} \\ &+ \beta_8 \ \text{FDI} * \text{CPIA}_{it} + \beta_9 \text{ODA} * \text{CPIA}_{it} + \ \epsilon_{it} \end{aligned}$$

Where;

HDI<sub>it</sub> is Human Development Index, FDI<sub>it</sub> is foreign direct investment, REM<sub>it</sub> is remittances, PG<sub>it</sub> is the population growth rate, CPIA<sub>it</sub> is a proxy for institutional quality, GEEH<sub>it</sub> is government expenditure on education and health. The interaction terms for remittances, foreign direct investment, official development assistance, and institutional quality are REM\*CPIA, FDI\*CPIA and ODA\*CPIA, respectively (All interaction terms were created by mean-centered variables to minimize multicollinearity).  $u_i$  is a time invariant country-specific fixed effect term,  $\varepsilon_{it}$  is error term;  $\beta_0$  is the intercept,  $\beta_1$  ...,  $\beta_n$  are beta-coefficients, "i" is cross-section units (countries) and "t" is the study period (2009 to 2019). The definition and measurement of the variables are shown in Appendix I.



#### 5. Findings and discussion

#### 5.1. Diagnostic tests

Considering that different panel unit root tests make different assumptions regarding the cross correlation among the errors in the panel, the empirical study started with the test for cross-sectional independence. Unit root tests may yield false results if there is a substantial amount of cross-section dependence and it is ignored. Three cross-sectional independence tests were used in the study: the Friedman (1937) statistic, the test statistic proposed by Frees (1995) and Pesaran's (2004) cross-sectional dependence (CD) test. The general null hypothesis H0: cross-sectional independence and the alternative hypothesis H1: cross-sectional dependence are both applied to the three tests. We fail to reject the null hypothesis at a 1% level based on the results of the three tests provided in Table 1, as all of the p-values are greater than 5%.

To test for stationarity, the study employed the Levin-Lin-Chu (LLC), Im, Pesaran and Shin (IPS), and Fisher-type unit root (Based on augmented Dickey–Fuller tests) (Fisher-ADF) tests. The results reported in Table 2 confirm that the data is stationary. To check for multicollinearity, the study used the variance inflation factor (VIF; Docquier et al., 2007). The stronger the correlation, the higher the VIF value. If the VIF value is greater than 10, then there may be a multicollinearity problem (Enders, 2004; Gujarati, 2004). Given that the maximum VIF value of 1.71, as shown by the results in Table 2, there is no multicollinearity problem with the studied variables.

#### 5.2. Descriptive statistics

The descriptive statistics for the research variable are presented in Table 3. The mean Human Development Index (HDI) was 0.489 (minimum = 0.304 and maximum = 0.796; standard deviation = 0.079). According to the UNDP, HDI of over 0.8 is considered high, while 0.5–0.08 is medium. Conversely, the HDI of 0.0–0.5 is low. Therefore, the mean of 0.489 suggests that the selected SSA countries have low human capital development. Remittances had a mean value of 3.728 (minimum = 0.073 and maximum = 16.285; standard deviation = 3.543). Official Development Assistance (ODA) had a mean value of 7.381 (minimum = 0.073 and maximum = 21.142; standard deviation = 5.127). The mean foreign direct investment (FDI) in the SSA was 3.988 (minimum = 0.238 and maximum = 19.456; standard deviation = 3.421). Furthermore, the mean CPIA

Table 1. Cross sectional Independence						
Test	Pesaran		Friedman		Frees	
	CD test	ρ-value	CD	ρ-value	CD (Q)	ρ-value
RE Model	0.467	0.6406	13.738	0.9987	2.629	0.465
FE Model	0.600	0.5487	16.604	0.995	2.958	0.465

FE and RE stand for fixed effects and random effects models, respectively. p-value significance at 1% level.

Table 2. Analysis of unit root and multicollinearity					
Variable	LLC	Fisher-ADF	IPS	VIF	1/VIF
HDI	-9.995(0.000)	138.099(0.000)	-3.043(0.000)	-	-
REM	-67.061(0.000)	140.992(0.000)	-2.363(0.001)	1.51	0.661706
FDI	-6.846(0.000)	195.038(0.000)	-2.904(0.002)	1.42	0.706627
ODA	-7.010(0.000)	100.793(0.000)	-2.961(0.000)	1.71	0.585973
CPIA	-6.943(0.000)	101.567(0.000)	-1.893(0.003)	1.07	0.937846
GEEH	-9.299(0.000)	131.168(0.000)	-2.042(0.002)	1.04	0.959957
PGR	-17.524(0.000)	248.964(0.000)	-2.459(0.001)	1.41	0.710816

Note. p-values are in parentheses



Table 3. Summary statistics for the research variables					
Variable	Obs	Mean	Std. Dev.	Min	Max
Human development index (HDI)	374	0.489	0.079	0.304	0.796
Remittances(% GDP)	374	3.728	3.543	0.073	16.285
Foreign direct investment(% GDP)	374	4.416	4.082	0.238	20.971
Official development assistance(% GDP)	374	7.381	4.951	0.073	21.142
CPIA	374	2.763	0.578	1.000	4.000
Government expenditure on education and health(%GDP)	374	5.618	1.357	2.762	10.930
Population growth rate (% annual)	374	2.567	0.642	0.182	3.8471

(institutional quality) was 2.763 (minimum = 1.000 and maximum = 4.000; standard deviation = 0.722), while the average government expenditure on education and health care was 5.618 (minimum = 1.357 and maximum = 10.930; standard deviation 1.357). The average population growth rate for the region was 2.567 (minimum = 0.182 and maximum = 3.871; standard deviation = 0.642).

#### 5.3. Regression analysis

The study tested the hypotheses using the regression findings from the system GMM. The Sargan/ Hansen test for overidentifying restrictions and the second-order serial correlation test AR (2) test were initially carried out to assess the issue of endogeneity and the validity of the instruments. The GMM system technique was used to apply the Sargan and Hansen test of overidentifying restrictions as suggested by Arellano and Bond (1991) and Hansen (1982). Given that the probability values for the Sagan/Hansen test and the AR (2) test were not significant at 5%, confirm that there was no second-order serial correlation. Table 4 confirms that the instruments are valid.

The results presented in Model I indicate that remittances have a significant and positive effect on human capital development ( $\beta$  = 0.004,  $\rho$  < 0.05); thus, H1 is supported. The findings are consistent with those of Azam and Raza (2016), Borja (2020), and Umar (2022) but contradict Hassan et al. (2013) who reported a negative effect. Remittances alleviate the migrant's household income constraints, hence the receiving household is capable of investing more in education and healthcare. The increased budget for education would also mean more years of schooling at both the secondary and tertiary school levels, which in the long run improves a country's stock of human capital. In addition, migrants are more likely to take children left behind to private schools that offer better education compared to public schools. The findings emphasize the necessity of targeted regulations to lower the cost of foreign money transfers and the necessity of enhancing channels for efficient remittance utilization. The receiving countries must also promote an environment that encourages the investment of this capital flow in areas connected to the advancement of human capital.

The results further show that the association between foreign direct investment and human capital development is significant and positive ( $\beta$  = 0.003,  $\rho$  < 0.05); hence, H2 is supported. The findings are consistent with those of Zhuang (2017) and Henok and Kaulihowa (2021) but differ



Table 4. Regression results					
HDI	Model 1.	Model 2.	Model.3	Model 4	Model. 5
L1.	.862(0.018)**	.859(0.018)**	.853(0.018)**	.853(0.018)**	.850(0.018)**
CONSTANT	025(0.015)	.017(0.015)	.021(0.015)	.020(0.015)	.021(0.015)
GEEH	.002(0.001)**	.002(0.001)**	.002(0.001)**	.002(0.001)**	.002(0.001)**
PGR	004(0.002)**	004(0.002)**	005(0.002)**	004(0.002)**	004(0.002)**
REM	.004(0.001)**	.004(0.001)**	.003(0.001)**	.003(0.001)**	.003(0.001)**
FDI	.003(0.001)**	.002(0.001)**	.003(0.001)**	.003(0.002)**	.003(0.001)**
ODA	007(0.003)**	008(0.003)**	007(0.003)**	006(0.003)**	005(0.003)**
CPIA	-	.004(0.002)**	.005(0.002)**	.005(0.002)**	.004(0.002)**
REM*CPIA	-	-	.006(0.003)**	.006(0.003)**	.006(0.002)**
FDI*CPIA	-	-	-	.007(0.002)**	.007(0.002)**
ODA*CPIA	-	-	-	-	.001(0.001)
No.instruments	51	52	53	54	55
No. groups	34	34	34	34	34
Post-estimation analysis					
Arellano-Bond:AR(1)		0.052			
Arellano-Bond:AR(2)		0.128			
Sargan test (p-val)		0.445			
Hansen test (p-val)		0.740			

Notes: system-GMM estimations for dynamic panel-data models. HDI, human development index; GEEH, Government expenditure on education and health; PG, population growth rate; REM, remittances; FDI, foreign direct investment; ODA, official development assistance; CPIA, country policy and institutional assessment; Standard error in the parenthesis; \*\*Significant at 5 per cent level

from those of Checchi et al. (2007) who reported a negative relationship. Meanwhile, Yildirim and Tosuner (2014), Sun and He (2014) found no relationship. FDI may contribute to human capital development through a direct effect of the upgraded skill level of the workforce and indirect effects through improved healthcare and spillover effect. Conversely, official development assistance had a significantly negative effect on human capital development in sub-Saharan Africa ( $\beta = -0.007$ ,  $\rho < 0.05$ ); thus, hypothesis H3 is rejected. These findings are supported by previous studies (Asongu, 2014; Lohani, 2004). The results conflict with earlier studies that reported a positive relationship (S. A. Asongu & Tchamyou, 2019; Moe, 2008; Yogo, 2017). The effectiveness of foreign aid in human development and economic growth is largely a controversial topic. The negative effect of ODA on human capital development can be attributed to misappropriation of aid meant for education and health programs. Additionally, while foreign aid may broaden access to healthcare and education, an unfavorable governmental environment in the recipient nation may compromise the quality of these services. Furthermore, foreign aid breeds dependency, which could result in lower domestic spending on the health and education sectors.

The regression results shown in Model 3, Table 4, demonstrate that institutional quality positively influences the growth of human capital ( $\beta$  = 0.004,  $\rho$  < 0.05), supporting hypothesis H4. The results support those of prior studies (Bertoni et al., 2019; Fagbemi et al., 2022; Koppensteiner & Menezes, 2021; Ouedraogo et al., 2022). The findings demonstrate that raising institutional quality encourages accessibility to healthcare, education, and other public goods that support human capital development. Therefore, the most crucial factors that promote the development of human capital are the efficacy of the government in preventing corruption, maintaining political stability and upholding the rule of law. The findings imply that promoting institutional quality dimensions is essential to enhancing the development of human capital in sub-Saharan Africa.



From Models 3–5, the interaction terms are added stepwise to the regression model. The moderating hypotheses (H5a-c) are tested using the pooled model (Model 5). Based on Model 5, Table 4, hypotheses H5a and H5b are validated. The coefficients of all interaction terms of remittances and institutional quality (REM \*CPIA) have a positive and significant effect on human capital development ( $\beta$  = 0.006,  $\rho$  < 0.05). The results also show the positive and significant impacts of interaction terms of FDI with institutional quality (FDI \*CPIA) on human capital development ( $\beta$  = 0.007,  $\rho$  < 0.05). Hence, the study's findings confirm that a high quality of institutions system strengthens the effect of remittances and FDI on the dimensions of human capital developments such as education and health. Prior studies have also reported that the developmental impact of external finance on economic growth is more pronounced for countries with sound economic policies and stronger institutions of governance (Catrinescu et al., 2009; Chitambara, 2019; Le, 2009; Zghidi et al., 2018). Empirical findings of the study suggest that apart from putting appropriate and effective macroeconomic policies to attract more international remittances and FDI to promote human capital development, it is important to promote good governance.

The findings further indicate that the interaction of official development assistance and institutional quality (ODA\*CPIA) is positive but insignificant ( $\beta$  = 001,  $\rho$  > 0.05); therefore, H5c is rejected. These results imply that sound institutions may not avert the adverse effects of foreign aid on human capital development. Foreign aid may deteriorate governance standards of the recipient countries. For instance, prior studies report a positive relationship between foreign aid and corruption (Asongu, 2012; S. Asongu & Jellal, 2013). The collaboration between weak institutions and development assistance that breeds corruption negatively affects human capital development.

#### 6. Conclusions

This study examined whether institutional quality moderates the relationship between foreign capital flows and human capital development in sub-Saharan Africa. A sample of 34 countries and data for 2009 to 2019 was used. The data was extracted from the WDI of the World Bank, UNDP and the OECD. The findings showed that remittances, foreign direct investment and institutional quality are positive drivers of human capital development in SSA. However, ODA has an adverse effect on human capital development. The study's findings further revealed that institutional quality moderated the relationship between remittances, FDI and human capital. However, the interaction between the quality of institutions and aid is insignificant. On the basis of these findings several policy implications can be drawn. First, policymakers interested in enhancing the benefits of foreign capital flow among developing countries can leverage remittances and FDI to increase human capital development. Second, since remittances are usually transferred through informal channels such as friends or family members traveling abroad policy interventions that can encourage the transmission of these capital flows through conventional financial systems are necessary. Therefore, this study argues that sub-Saharan African countries can strategically harness the contributions of remittances and FDI to increase their level of human capital development. To do this, SSA nations should ensure effective and reliable transfers of external capital by lowering the cost of transfers and improving the quality of institutions. At the same time, the government should leverage remittances for human capital development through improved financial literacy and expanded financial inclusion, particularly in rural areas where majorities of households receiving remittances live. Third, the positive interaction effect of institutional quality and foreign direct investment emphasizes the need for friendly policies to encourage foreign investors to invest in priority areas of human capital development. Fourth, ODA functions effectively in nations with sound institutions. However, corruption or improper administration of foreign aid received and poor governance are typical of developing nations (like those that make up SSA). As a result, ODA funds end up being misappropriated instead of carrying out investment initiatives that benefit the general populace. Therefore, to promote socioeconomic development, development partners and international organizations should help developing nations create solid legal and political institutions. Finally, there is a need for improved coordination and management of donor funds for aid to increase the positive externalities.

Despite the contributions of this study, there are several limitations. First, human capital development is measured by HDI that is composite index of life expectancy, education and income. Therefore, future studies can consider other dimensions of human development like poverty, gender disparity, inequality and health status. Second, the data used in this study uses the aggregated amounts of remittances, FDIs and aid. Hence, it would be important for future studies to explore the specific effect of these foreign capitals on human capital development by classifying FDI and aid by purpose and type. In addition, the association between remittances, institutional quality and human capital development may be studied at household level. Third, the study is limited to SSA countries, and it may be difficult to replicate the studies in regions receiving low foreign capital flows and with strong institutions.

#### Funding

The authors have no funding to report.

#### **Author details**

Peter Nderitu Githaiga<sup>1</sup> E-mail: nderitugithaiga@mu.ac.ke

ORCID ID: http://orcid.org/0000-0003-0772-7854 Andrew Wenani Kilong'i<sup>2</sup>

- <sup>1</sup> Department of Accounting and Finance, Moi University, Eldoret, Kenya.
- <sup>2</sup> Department of History, Political science and Public Administration, Moi University, Eldoret, Kenya.

#### Disclosure statement

No potential conflict of interest was reported by the author(s).

#### Citation information

Cite this article as: Foreign capital flow, institutional quality and human capital development in sub-Saharan Africa, Peter Nderitu Githaiga & Andrew Wenani Kilong'i, Cogent Economics & Finance (2023), 11: 2162689.

#### References

- Acemoglu, D., Johnson, S., & Robinson, J. A. (2005). Institutions as a fundamental cause of long-run growth. *Handbook of Economic Growth*, 1, 385–472. https://doi.org/10.1016/S1574-0684(05)01006-3
- Adams, S., Klobodu, E. K. M., & Lamptey, R. O. (2017). The effects of capital flows on economic growth in Senegal. *Margin: The Journal of Applied Economic Research*, 11(2), 121–142. https://doi.org/10.1177/0973801016687869
- Affandi, Y., Anugrah, D. F., & Bary, P. (2019). Human capital and economic growth across regions:
  A case study in Indonesia. *Eurasian Economic Review*, *9*(3), 331–347. https://doi.org/10.1007/s40822-018-0114-4
- Ahmed, J., Mughal, M., & Martínez-Zarzoso, I. (2021). Sending money home: Transaction cost and remittances to developing countries. *The World Economy*, 44 (8), 2433–2459. https://doi.org/10.1111/twec.13110
- Ajide, K. B., & Raheem, I. D. (2016). The institutional quality impact on remittances in the ECOWAS subregion. *African Development Review*, 28(4), 462–481. https://doi.org/10.1111/1467-8268.12224
- Arellano, M., & Bond, S. (1991). Some tests of specification for panel data: Monte Carlo evidence and an application to employment equations. *The Review of Economic Studies*, 58(2), 277–297. https://doi.org/10.2307/2297968
- Arellano, M., & Bover, O. (1995). Another look at the instrumental variable estimation of error-components models. *Journal of Econometrics*, 68(1), 29–51. https://doi.org/10.1016/0304-4076(94)01642-D
- Asiama, J. P., & Quartey, P. (2009). Foreign aid and the human development indicators in Sub-Saharan

- Africa. Journal of Developing Societies, 25(1), 57–83. https://doi.org/10.1177/0169796X0902500103
- Asongu, S. (2012). Government Quality Determinants of Stock Market Performance in African Countries. Journal of African Business, 13(3), 183–199. https://doi.org/10.1080/15228916.2012.727744
- Asongu, S. (2014). Development thresholds of foreign aid effectiveness in Africa. *International Journal of Social Economics*, 41(11), 1131–1155. https://doi.org/10. 1108/IJSE-01-2013-0014
- Asongu, S., & Jellal, M. (2013). On the Channels of Foreign Aid to Corruption. *Economics Bulletin*, 33(3), 2191– 2201. https://dx.doi.org/10.2139/ssrn.2493353
- Asongu, S. A., & Nwachukwu, J. C. (2016). Foreign aid and governance in Africa. *International Review of Applied Economics*, 30(1), 69–88.
- Asongu, S. A., & Tchamyou, V. S. (2019). Foreign aid, education and lifelong learning in Africa. *Journal of* the Knowledge Economy, 10(1), 126–146. https://doi. org/10.1007/s13132-017-0449-1
- Azam, M., & Raza, S. A. (2016). Do workers' remittances boost human capital development? *The Pakistan Development Review*, 55(2), 123–149. https://doi.org/10.30541/v55i2pp.123-149
- Barua, S. (2021). Human Capital, Economic Growth, and Sustainable Development Goals: An Evaluation of Emerging Economies. In *The Dynamics of Intellectual* Capital in Current Era (pp. 129–148). Springer.
- Bénassy-Quéré, A., Coupet, M., & Mayer, T. (2007). Institutional determinants of foreign direct investment. World Economy, 30(5), 764–782. https:// doi.org/10.1111/j.1467-9701.2007.01022.x
- Bertoni, E., Di Maio, M., Molini, V., & Nistico, R. (2019). Education is forbidden: The effect of the Boko Haram conflict on education in North-East Nigeria. *Journal of Development Economics*, 141, 102249. https://doi. org/10.1016/j.jdeveco.2018.06.007
- Bhattacharya, M., Inekwe, J., & Paramati, S. R. (2018). Remittances and financial development: Empirical evidence from heterogeneous panel of countries. Applied Economics, 50(38), 4099–4112. https://doi. org/10.1080/00036846.2018.1441513
- Bloom, D. E., Canning, D., Kotschy, R., Prettner, K., & Schünemann, J. J. (2019). Health and economic growth: Reconciling the micro and macro evidence. National Bureau of Economic Research. No. w26003.
- Blundell, R., & Bond, S. (1998). Initial conditions and moment restrictions in dynamic panel data models. Journal of Econometrics, 87(1), 115–143. https://doi.org/10.1016/S0304-4076(98)00009-8
- Borja, K. (2014). Social capital, remittances and growth. European Journal of Development Research, 26(5), 574–596.
- Borja, K. (2017). Corruption indicators, foreign capital, and economic growth in developing countries. *Journal of Developing Areas*, 51(4), 95–107.

- Borja, K. (2020). Remittances, corruption, and human development in Latin America. Studies in Comparative International Development, 55(3), 305–327. https://doi.org/10.1007/s12116-020-09299-1
- Boucekkine, R., Martínez, B., & Ruiz-Tamarit, J. R. (2013). Growth vs. level effect of population change on economic development: An inspection into human-capital-related mechanisms. *Journal of Mathematical Economics*, 49(4), 312–334. https://doi.org/10.1016/j.jmateco.2013.04.001
- Buchanan, B. G., Le, Q. V., & Rishi, M. (2012). Foreign direct investment and institutional quality: Some empirical evidence. *International Review of Financial Analysis*, 21, 81–89. https://doi.org/10.1016/j.irfa.2011.10.001
- Buevich, A. P., Varvus, S. A., & Terskaya, G. A. (2019, May). Investments in human capital as a key factor of sustainable economic development. In *Institute of Scientific Communications Conference* (pp. 397–406). Springer, Cham.
- Burnside, C., & Dollar, D. (1997). Aid spurs growth-in a sound policy environment. *Finance & Development*, 34(004).
- Catrinescu, N., Leon-Ledesma, M., Piracha, M., & Quillin, B. (2009). Remittances, institutions, and economic growth. *World Development*, 37(1), 81–92. https://doi.org/10.1016/j.worlddev.2008.02.004
- Checchi, D., De Simone, G., & Faini, R. (2007). Skilled migration, FDI and human capital investment (No. 2795). IZA Discussion Papers.
- Chitambara, P. (2019). Remittances, institutions and growth in Africa. *International Migration*, 57(5), 56–70. https://doi.org/10.1111/imig.12542
- Collier, P., & Dollar, D. (2002). Aid allocation and poverty reduction. European economic review, 46(8), 1475–
- Daniele, V., & Marani, U. (2011). Organized crime, the quality of local institutions and FDI in Italy: A panel data analysis. European Journal of Political Economy, 27(1), 132–142.
- Darley, W. K. (2012). Increasing Sub-Saharan Africa's share of foreign direct investment: Public policy challenges, strategies, and implications. *Journal of African Business*, 13(1), 62–69. https://doi.org/10. 1080/15228916.2012.657981
- Di Maio, M., & Nandi, T. K. (2013). The effect of the Israeli– Palestinian conflict on child labor and school attendance in the West Bank. *Journal of Development Economics*, 100(1), 107–116. https://doi.org/10.1016/ i.ideveco.2012.08.007
- Dissou, Y., Didic, S., & Yakautsava, T. (2016). Government spending on education, human capital accumulation, and growth. *Economic Modelling*, 58, 9–21. https://doi.org/10.1016/j.econmod.2016.04.015
- Docquier, F., Lohest, O., & Marfouk, A. (2007). Brain drain in developing countries. *The World Bank Economic Review*, 21(2), 193–218. https://doi.org/10.1093/ wber/lhm008
- Enders, W. (2004). Applied econometric time series 2nd edition. New York: John Willey & Sons. *Technometrics*, 46(2), 264.
- Fagbemi, F., & Osinubi, T. T. (2020). Leveraging foreign direct investment for sustainability: An approach to sustainable human development in Nigeria. Resources, Environment and Sustainability, 2, 100005. https://doi.org/10.1016/j.resenv.2020.100005
- Fagbemi, F., Osinubi, T. T., Nzeribe, G. E., & Bankole, T. O. (2022). Human Capital Development Challenge: Why Corruption Eradication is a Panacea in Nigeria. Journal of Development Policy and Practice, 7(2), 180– 205. https://doi.org/10.1177/24551333221090312

- FAO and ECA. (2018). Regional Overview of Food Security and Nutrition. Addressing the threat from climate variability and extremes for food security and nutrition. Accra.
- Fertig, M., Schmidt, C. M., & Sinning, M. G. (2009). The impact of demographic change on human capital accumulation. *Labour Economics*, 16(6), 659–668. https://doi.org/10.1016/j.labeco.2009.08.005
- Frees, E. W. (1995). Assessing cross-sectional correlation in panel data. *Journal of Econometrics*, 69(2), 393–414. https://doi.org/10.1016/0304-4076(94) 01658-M
- Friedman, M. (1937). The use of ranks to avoid the assumption of normality implicit in the analysis of variance. *Journal of the American Statistical Association*, 32(200), 675–701. https://doi.org/10. 1080/01621459.1937.10503522
- Gujarati, D. (2004). Basic econometrics. Tata McGraw-Hill Publishing Company Ltd.
- Hansen, L. P. (1982). Large sample properties of generalized method of moments estimators. Econometrica: Journal of the Econometric Society, 50(4), 1029–1054.
- Hassan, M. U., Mehmood, H., & Hassan, M. S. (2013). Consequences of Worker's Remittances on Human Capital: An In-Depth Investigation for a Case of Pakistan. Middle-East Journal of Scientific Research, 14(3), 443–452. https://escholar.umt.edu.pk:8080/ ispui/handle/123456789/864
- Henok, W., & Kaulihowa, T. (2021). The impact of FDI on human capital development in SACU countries. International Journal of Social Economics, 49(2), 268–279. https://doi.org/10.1108/IJSE-02-2021-0123
- Heyse, P., Mahieu, R., & Timmerman, C. (2015). The migration trajectories of Russian and Ukrainian women in Belgium. In C. Timmerman, M. Martiniello, & A. Rea (Eds.), New dynamics in female migration and integration (pp. 80–113). Routledge.
- Idike, A. N., Ukeje, I. O., Ogbulu, U., Aloh, J. N., Obasi, V. U., Nwachukwu, K., . . . Ejem, E. N. (2021). The Practice of Human Capital Development Process and Poverty Reduction: Consequences for Sustainable Development Goals in Ebonyi State, Nigeria. Public Organization Review, 21(2), 263–280. https://doi.org/ 10.1007/s11115-020-00482-5
- Jhinger, M. (2005). The Economic of Development and Planning. Vinade Publication.
- Kar, S. (2013). Exploring the causal link between FDI and human capital development in India. *Decision*, 40(1), 3–13. https://doi.org/10.1007/s40622-013-0001-5
- Kathavate, J., & Mallik, G. (2012). The impact of the Interaction between institutional quality and aid volatility on growth: Theory and evidence. *Economic Modelling*, 29(3), 716–724. https://doi.org/10.1016/j.econmod.2012.01.020
- Katoka, B., & Kwon, H. J. (2018). Business Regulations and Foreign Direct Investment in Sub-Saharan Africa: Implications for Regulatory Reform. In U. Efobi & S. Asongu (Eds.), Financing Sustainable Development in Africa (pp. 63–91). Palgrave Macmillan.
- Kaulihowa, T., & Adjasi, C. (2019). Non-linearity of FDI and human capital development in Africa. *Transnational Corporations Review*, 11(2), 133–142. https://doi.org/ 10.1080/19186444.2019.1635734
- Kheng, V., Sun, S., & Anwar, S. (2017). Foreign direct investment and human capital in developing countries: A panel data approach. *Economic Change and Restructuring*, 50(4), 341–365. https://doi.org/10. 1007/s10644-016-9191-0
- Koppensteiner, M. F., & Menezes, L. (2021). Violence and human capital investments. *Journal of Labor*



- Economics, 39(3), 787-823. https://doi.org/10.1086/711001
- Kosack, S. (2003). Effective aid: How democracy allows development aid to improve the quality of life. World Development, 31(1), 1–22. https://doi.org/10.1016/S0305-750X(02)00177-8
- Kroeger, A., & Anderson, K. H. (2014). Remittances and the human capital of children: New evidence from Kyrgyzstan during revolution and financial crisis, 2005–2009. *Journal of Comparative Economics*, 42(3), 770–785. https://doi.org/10.1016/j.jce.2013.06.001
- Kulkami, V. S. (2014). Globalization of remittances in India: Towards a sociological perspective. In M. Rahman., T. Yong. & M. Ullah (Eds.), *Migrant* Remittances in South Asia (pp. 192–217). Palgrave
- La Porta, R., Lopez-de-silanes, F., Shleifer, A., & Vishny, R. W. (1997). Legal determinants of external finance. *The Journal of Finance*, 52(3), 1131–1150. https://doi.org/10.1111/j.1540-6261.1997.tb02727.x
- Lartey, E. K., & Mengova, E. (2016). Does institutional quality in developing countries affect remittances? The Journal of Developing Areas, 50(1), 59–76. https://doi.org/10.1353/jda.2016.0008
- Le, T. (2009). Trade, remittances, institutions, and economic growth. *International Economic Journal*, 23(3), 391–408. https://doi.org/10.1080/ 10168730903119443
- Lee, E., Jung, K., & Sul, J. (2019). Searching for the various effects of subprograms in official development assistance on human development across 15 Asian countries: Panel regression and fuzzy set approaches. Sustainability, 11(4), 1152. https://doi.org/10.3390/su11041152
- Lohani, S. (2004). Effect of foreign aid on development: Does more money bring more development? Illinois Wesleyan University. Department of Economics.
- Lucas, R. E. (1988). On the mechanics of economic development. Journal of Monetary Economics, 22(1), 3-42. https://doi.org/10.1016/0304-3932(88)90168-7
- Lucas, R. E., & Stark, O. (1985). Motivations to remit: Evidence from Botswana. *Journal of Political Economy*, 93(5), 901–918. https://doi.org/10.1086/ 261341
- Matousek, R., & Tzeremes, N. G. (2021). The asymmetric impact of human capital on economic growth. Empirical Economics, 60(3), 1309–1334. https://doi.org/10.1007/s00181-019-01789-z
- Mincer, J. (1996). Economic development, growth of human capital, and the dynamics of the wage structure. *Journal of Economic Growth*, 1(1), 29–48. https://doi.org/10.1007/BF00163341
- Moe, T. L. (2008). An empirical investigation of relationships between official development assistance (ODA) and human and educational development. *Economics*, 35(3), 202–221. https://doi.org/10.1108/03068290810847879
- Moore, M. D., & Daday, J. (2010). Barriers to human capital development: Case studies in Swaziland, Cameroon and Kenya. Africa Education Review, 7(2), 283–304. https://doi.org/10.1080/18146627.2010.515418
- Munteanu, A. C. (2015). Knowledge spillovers of FDI. Procedia Economics and Finance, 32, 1093–1099. https://doi.org/10.1016/S2212-5671(15)01573-7
- Musibau, H. O., Yusuf, A. H., & Gold, K. L. (2019). Endogenous specification of foreign capital inflows, human capital development and economic growth: A study of pool mean group. *International Journal of Social Economics*, 46(3), 454–472. https://doi.org/10. 1108/IJSE-04-2018-0168

- North, D. C. (1990). Institutions, institutional change and economic performance. Cambridge university press.
- OECD. (2005). Glossary of foreign direct investment terms and definitions. https://www.oecd.org/daf/inv/invest ment-policy/2487495.pdf
- OECD. (2019). Development Aid at a Glance Statistics by Region https://www.oecd.org/dac/financingsustainable-development/development-finance-data /Africa-Development-Aid-at-a-Glance-2019.pdf
- OECD. (2021).https://www.oecd.org/development/finan cing-sustainable-development/development-finance-standards/official-development-assistance.htm.

  Accessed on 11th December, 2021
- Olopade, B. C., Okodua, H., Oladosun, M., & Asaleye, A. J. (2019). Human capital and poverty reduction in OPEC member-countries. *Heliyon*, 5(8), e02279. https://doi.org/10.1016/j.heliyon.2019.e02279
- Organisation for Economic Cooperation and Development. (2012). Official Development Assistance (ODA). OECD Publishing. Retrieved from http://www.oecd.org/dac/stats/documentupload/ What-is-ODA.pdf
- Ortega, B., Casquero, A., & Sanjuán, J. (2016). Corruption and convergence in human development: Evidence from 69 countries during 1990–2012. Social Indicators Research, 127(2), 691–719. https://doi.org/10.1007/s11205-015-0968-8
- Ouedraogo, I., Tabi, H. N., Ondoa, H. A., & Jiya, A. N. (2022). Institutional quality and human capital development in Africa. *Economic Systems*, 46(1), 100937. https://doi.org/10.1016/j.ecosys.2021. 100937
- Penrose, E. (1959). The Theory of the Growth of the Firm. Oxford university press.
- Pesaran, M. H. (2004). General diagnostic tests for cross section dependence in panels (IZA Discussion. Erişim adresi: The Institute for the Study of Labor Discussion website:
- Petty, W. (1769). A Treatise of Taxes and Contributions. History of Economic Thought Books. https://www.har gaden.com/enda/wp-content/petty\_taxescontribu tions.pdf
- Ponce, J., Olivié Aldasoro, L., & Onofa, M. (2008). Remittances for Development?: A Case Study of the Impact of Remittances on Human Development in Ecuador. Elcano Newsletter, 48, 33.
- Rodrik, D., Subramanian, A., & Trebbi, F. (2004).
  Institutions rule: The primacy of institutions over geography and integration in economic development. *Journal of Economic Growth*, 9(2), 131–165. https://doi.org/10.1023/B:JOEG. 0000031425.72248.85
- Romer, P. (1990a). Endogenous technological changes. *Journal of Political Economy*, *98*(5, Part 2), S71–S102. https://doi.org/10.1086/261725
- Romer, P. (1990b). Human capital and growth: Theory and evidence. In *Carnegie-Rochester Conference Series on Public Policy* (Vol. 32, No. 1, pp. 251–286). Elsevier.
- Salas, V. B. (2014). International remittances and human capital formation. *World Development*, 59, 224–237. https://doi.org/10.1016/j.worlddev.2014.01.035
- Sarwar, A., Hassan, M., & Mahmood, T. (2015). Foreign aid and governance in Pakistan. Pakistan Economic and Social Review, 149–176. https://www.jstor.org/stable/ 26153255
- Schultz, T. W. (1961). Investment in human capital. *The American Economic Review*, 51(1), 1–17. https://www.jstor.org/stable/1818907
- Sethi, N., Bhujabal, P., Das, A., & Sucharita, S. (2019). Foreign aid and growth nexus: Empirical evidence



- from India and Sri Lanka. Economic Analysis and Policy, 64, 1–12.
- Shah, S. H., Ahmad, M. H., & Ahmed, Q. M. (2016). The nexus between sectoral FDI and institutional quality: Empirical evidence from Pakistan. Applied Economics, 48(17), 1591–1601. https://doi.org/10.1080/ 00036846.2015.1103039
- Shirazi, N. S., Mannap, T. A. A., & Ali, M. (2009).
  Effectiveness of foreign aid and human development.
  The Pakistan Development Review, 853–862. https://
  EconPapers.repec.org/RePEc:pid:journl:v:48:y:2009:
  i:4:p:853-862
- Smith, A. (1904). An inquiry into the nature and causes of the wealth of nations (5th) ed., E. Cannan, Ed.). Methuen. Retrieved September 20, 2021, from http:// www.econlib.org/library/Smith/smWN.html (Original work published 1776)
- Suliman, A. H., & Mollick, A. V. (2009). Human capital development, war and foreign direct investment in sub-Saharan Africa. Oxford Development Studies, 37 (1), 47–61. https://doi.org/10.1080/ 13600810802660828
- Sun, M., & He, Q. (2014). Does foreign direct investment promote human capital accumulation? The role of gradual financial liberalization. *Emerging Markets Finance and Trade*, 50(4), 163–175. https://doi.org/10. 2753/REE1540-496X500410
- Torruam, J. T., Chiawa, M. A., & Abur, C. C. (2014). Cointegration analysis of public expenditure on tertiary education and economic growth in Nigeria. *CBN Journal of Applied Statistics (JAS)*, 5(2), 7. https://dc.cbn.gov.ng/jas/vol5/iss2/7
- Umar, M. (2022). Remittances, institutions and human development in Sub-Saharan Africa. *Journal of Economics and Development*, 24(2), 142–157. https:// doi.org/10.1108/JED-03-2021-0041
- UNAIDS. (2018). Miles To Go: The Response to HIV in Eastern and Southern Africa. UNAIDS Joint United Nations Programme on HIV/AIDS.
- UNCTAD. (2018). The Least Developed Countries Report 2009: Entrepreneurship for structural transformation: Beyond business as usual (United Nations publication. Sales No. E.18.II.D.6)
- UNCTAD. (2021). https://unctad.org/news/global-foreigndirect-investment-fell-42-2020-outlook-remains-weak
- United nations Development Programme (UNDP) (2021). Human Development Index (HDI). Available at: http://www.hdr.undp.org/en/content/human-development-index-hdi (Accessed on April, 15).
- United Nations Development Program (UNDP). (2018).

  Human Development Indices and Indicators 2018

  Statistical Update. United Nations Development
  Programme. https://hdr.undp.org/system/files/documents//2018humandevelopmentstatisticalupdatepdf.pdf
- Wang, Q., Guo, J., & Li, R. (2022). Official development assistance and carbon emissions of recipient countries: A dynamic panel threshold analysis for low-and

- lower-middle-income countries. Sustainable Production and Consumption, 29, 158–170. https://doi.org/10.1016/j.spc.2021.09.015
- Weisbrod, B. A. (1962). Education and investment in human capital. *Journal of Political Economy*, 70(5, Part 2), 106–123. https://doi.org/10.1086/258728
- Wenzel, R. (2017). Learning for Purpose: Challenges and opportunities for human capital development in the social sector. In S. Tirmizi. & J. Vogelsang (Eds.), Leading and managing in the social sector (pp. 113–137). Springer.
- Word Bank. (2021). https://www.worldbank.org/en/news/ press-release/2021/11/17/remittance-flows-registerrobust-7-3-percent-growth-in-2021. Accessed on 11th December, 2021
- World Bank. (2019). World development indicator. 09/ November/2021 https://data.worldbank.org/indica tor/SP.DYN.LE00.IN?name desc=true
- World Health Organization. (2006). Meeting the MDG drinking water and sanitation target: The urban and rural challenge of the decade.
- Yang, X. (2020). Health expenditure, human capital, and economic growth: An empirical study of developing countries. International Journal of Health Economics and Management, 20(2), 163–176. https://doi.org/10.1007/s10754-019-09275-w
- Yao, Y. (2019). Does higher education expansion enhance productivity? *Journal of Macroeconomics*, 59, 169–194. https://doi.org/10.1016/j.jmacro.2018.11.009
- Yiheyis, Z., & Woldemariam, K. (2020). Remittances, Official Development Assistance, and Human Development in Africa: An Empirical Analysis. *Journal* of African Development, 21(2), 189–212. https://doi. org/10.5325/jafrideve.21.2.0189
- Yildirim, D. Ç., & Tosuner, Ö. (2014). The effects of FDI on human capital stock in Central Asian Turkic Republics. Eurasian Journal of Business and Economics, 7(14), 51–60. https://doi.org/10.17015/ ejbe.2014.014.03
- Yogo, T. U. (2017). Assessing the effectiveness of foreign aid in the education sector in Africa: The case of primary education. *African Development Review*, 29 (3), 389–402. https://doi.org/10.1111/1467-8268. 12276
- Zamani, Z., & Tayebi, S. K. (2022). Spillover effects of trade and foreign direct investment on economic growth: An implication for sustainable development. *Environment*, *Development and Sustainability*, 24(3), 3967–3981. https://doi.org/10.1007/s10668-021-01597-5
- Zghidi, N., Sghaier, I. M., & Abida, Z. (2018).
  Remittances, institutions, and economic growth in North African countries. *Journal of the Knowledge Economy*, 9(3), 804–821. https://doi.org/10.1007/s13132-016-0377-5
- Zhuang, H. (2017). The effect of foreign direct investment on human capital development in East Asia. *Journal* of the Asia Pacific Economy, 22(2), 195–211. https:// doi.org/10.1080/13547860.2016.1240321



#### APPENDIX I

Table A1. Description of var	
Variable	Definition
Human Capital Development	Human capital development is measured using the Human Development Index. The Human Development Index (HDI) is a summary measure of average achievement in key dimensions of human development: a long and healthy life, being knowledgeable and have a decent standard of living. The HDI is the geometric mean of normalized indices for each of the three dimensions UNDP (2021).
Foreign Direct Investment (FDI)	"Foreign direct investment refers to direct investment equity flows in an economy. It is the sum of equity capital, reinvestment of earnings, and other capital. Direct investment is a category of cross border investment associated with a resident in one economy having control or a significant degree of influence on the management of an enterprise that is resident in another economy. Ownership of 10 percent or more of the ordinary shares of voting stock is the criterion for determining the existence of a direct investment relationship. This series shows net outflows of investment from the reporting economy to the rest of the world. Data are in current U.S. dollar" (Word Bank, 2021).
Remittances	"Personal remittances comprise personal transfers and compensation of employees. Personal transfers consist of all current transfers in cash or in kind made or received by resident households to or from nonresident households. Personal transfers thus include all current transfers between resident and nonresident individuals.  Compensation of employees refers to the income of border, seasonal, and other short-term workers who are employed in an economy where they are not resident and of residents employed by nonresident entities. Data are the sum of two items defined in the sixth edition of the IMF's Balance of Payments Manual: personal transfers and compensation of employees" (Word Bank, 2021).
Official Development Assistance	"Net official development assistance (ODA) consists of disbursements of loans made on concessional terms (net of repayments of principal) and grants by official agencies of the members of the Development Assistance Committee (DAC), by multilateral institutions, and by non-DAC countries to promote economic development and welfare in countries and territories in the DAC list of ODA recipients. It includes loans with a grant element of at least 25 percent (calculated at a rate of discount of 10 percent). Data are in current U.S. dollars" (Word Bank, 2021).
Institutional Quality	"Transparency, accountability, and corruption in the public sector assess the extent to which the executive can be held accountable for its use of funds and for the results of its actions by the electorate and by the legislature and judiciary, and the extent to which public employees within the executive are required to account for administrative decisions, use of resources, and results obtained. The three main dimensions assessed here are the accountability of the executive to oversight institutions and of public employees for their performance, access of civil society to information on public affairs, and state capture by narrow vested interests." According to World Bank institutional quality is measured by CPIA which is an indicator of transparency, accountability, and corruption in the public sector rating (1 = low to 6 = high; Word Bank, 2021).
Population Growth	"Annual population growth rate for year t is the exponential rate of growth of midyear population from year t-1 to t, expressed as a percentage. Population is based on the de facto definition of population, which counts all residents regardless of legal status or citizenship" (Word Bank, 2021).
Government expenditure on education and health	A percentage which represents total expenditure on education and health (current and capital) undertaken by the State, over GDP. It includes government spending on educational institutions (both public and private), education administration, and subsidies for private entities (students/households and other private entities). Source: (Word Bank, 2021)





#### © 2023 The Author(s). This open access article is distributed under a Creative Commons Attribution (CC-BY) 4.0 license.

You are free to:

Share — copy and redistribute the material in any medium or format.

Adapt — remix, transform, and build upon the material for any purpose, even commercially.

The licensor cannot revoke these freedoms as long as you follow the license terms.

Under the following terms:



Attribution — You must give appropriate credit, provide a link to the license, and indicate if changes were made. You may do so in any reasonable manner, but not in any way that suggests the licensor endorses you or your use. No additional restrictions

 $You \ may \ not \ apply \ legal \ terms \ or \ technological \ measures \ that \ legally \ restrict \ others \ from \ doing \ anything \ the \ license \ permits.$ 

## Cogent Economics & Finance (ISSN: 2332-2039) is published by Cogent OA, part of Taylor & Francis Group. Publishing with Cogent OA ensures:

- Immediate, universal access to your article on publication
- · High visibility and discoverability via the Cogent OA website as well as Taylor & Francis Online
- Download and citation statistics for your article
- · Rapid online publication
- Input from, and dialog with, expert editors and editorial boards
- · Retention of full copyright of your article
- Guaranteed legacy preservation of your article
- Discounts and waivers for authors in developing regions

#### Submit your manuscript to a Cogent OA journal at www.CogentOA.com

